Author’s response to reviews

Title: FACTORS PREDICTING ONE-YEAR POST-SURGICAL MORTALITY AMONGST OLDER ASIAN PATIENTS UNDERGOING MODERATE TO MAJOR NON-CARDIAC SURGERY – A RETROSPECTIVE COHORT STUDY

Authors:

Lydia Liew (lydia_qn_liew@nuhs.edu.sg)
Wei Wei Teo (wei_wei_teo@nuhs.edu.sg)
Edwin Seet (seet.edwin.cp@ktph.com.sg)
Lyn Lean (lyn_li_lean@nuhs.edu.sg)
Ambika Paramasivan (ambika_paramasivan@nuhs.edu.sg)
Joanna Tan (joanna_tan@nuhs.edu.sg)
Irene Lim (irene_al_lim@nuhs.edu.sg)
Jiexun Wang (wang.jiexun@ktph.com.sg)
Lian Ti (anatilk@nus.edu.sg)

Version: 2 Date: 06 Oct 2019

Author’s response to reviews:

This is attached as a separate document titled 'Cover Letter for Revisions'.